Author's response to reviews

Title: Rapid semi-automated quantitative multiplex tandem PCR (MT-PCR) assays for the differential diagnosis of influenza-like illness

Authors:

Elektra Szewczuk (keith.stanley@ausdx.com.au)
Kiran Thapa (kiran.thapa@swahs.health.nsw.gov.au)
Terry Anninos (terry.anninos@imvs.sa.gov.au)
Kenneth McPhie (ken.mcphie@swahs.health.nsw.gov.au)
Geoff Higgins (geoff.Higgins@imvs.sa.gov.au)
Dominic E Dwyer (dominic.dwyer@wmi.usyd.edu.au)
Keith K Stanley (keith.stanley@ausdx.com.au)
Jonathan R Iredell (jiredell@usyd.edu.au)

Version: 5 Date: 28 March 2010

Author's response to reviews: see over
Dear colleagues

Thank you for accepting this revised MS, which has been reformatted as requested. Figure 1 has been named Figures 1-4, and the Tables have been appended after the References section. All authors have been circulated the document on several occasions and no additional changes requested.

Kind regards

Jon Iredell